Author's response to reviews

Title: Knowledge, attitudes and practice of healthcare ethics and law among doctors and nurses in Barbados

Authors:

Seetharaman Hariharan (uwi.hariharan@gmail.com)
Ramesh Jonnalagadda (rameshj@caribsurf.com)
Errol Walrond (erwalrond@caribsurf.com)
Harley Moseley (hslm2@caribsurf.com)

Version: 4 Date: 8 June 2006

Author's response to reviews:

Formatting changes required by the core-preaccept has been done to the manuscript